Author's response to reviews

Title: Systematic review of the use of honey as a wound dressing

Authors:

Owen A Moore (omoore@doctors.org.uk)
Lesley A Smith (lesley.smith@pru.ox.ac.uk)
Fiona Campbell (fiona.campbell@rcn.org.uk)
Kate Seer (kate.seers@rcn.org.uk)
Henry J McQuay (henry.mcquay@pru.ox.ac.uk)
R Andrew Moore (andrew.moore@pru.ox.ac.uk)

Version: 6 Date: 29 May 2001

Table 1 is now as .xls as supp1 (supplementary material).
Table 2 now Table 1
Text amended as appropriate